Network Programming August 26, 1996 1/1

Network Programming Term Project

A significant term project is required and will account for 20% of your grade. Each term project will be submitted in 3 stages:

1. Project proposal. A brief (1 or 2 page) description of your proposed project. Your term project must be approved. Proposals are due by November 11th.

2. Presentation to the class is required. The class will evaluate each presentation as part of the grading process. Scheduling of presentations will be on a first-come-first-served basis - no scheduling is possible until a proposal is accepted.

3. A project write-up (and source code if you do a programming project) are due by the last day of classes (Dec 9th).

• Projects can be a programming project, an in-depth case study, or anything else approved by the instructor.

• Term projects can be done individually or in groups of any size. The scope of the project should reflect the number of people participating.

• The term project should be a learning endeavour, not simply a display of your knowledge of topics discussed in class. The presentation is a way to pass along your new knowledge to the rest of the class.

• Be aggressive in selecting a project! I would like to see everyone try something significant (even if that means you can’t finish the project). Easy projects will be graded as such.

• I encourage everyone to discuss potential projects with the class. Since the class will participate in the grading of your project - it may be helpful to hear whether or not the class thinks your topic is interesting.